Author's response to reviews

Title: Effects of work-related factors on the breastfeeding behavior of working mothers in a Taiwanese semiconductor manufacturer: a cross-sectional survey

Authors:

Yi Chun Chen (yichun@tmu.edu.tw)
Ya-Chi Wu (ycwu@cde.org.tw)
Wei-Chu Chie (weichu@ntu.edu.tw)

Version: 6 Date: 16 June 2006

Author's response to reviews: see over
Dear Editor,

Thank you very much for your e-mail on June 15, 2006 of the comments for our manuscript 5893251118278401. We have revised our manuscript according to your comments:

1. Title: We have changed the title to sentence case by removing all unnecessary capitals (P.1).
2. Background of the abstract: We have changed the last sentence to past tense (P.2).
3. Section headings: We have changed the subheadings to sentence case by removing all unnecessary capitals.
4. Table titles: We have changed the titles and group names to sentence case by removing all unnecessary capitals.
5. Additional file: We have revised the related text according to your comments (P.13).

Thank you again for your kind attention and we are looking forward to hearing from you soon.

Sincerely yours,

Wei-Chu Chie, M.D., Ph.D.
Professor, School of Public Health and Graduate Institute of Preventive Medicine
College of Public Health, National Taiwan University
Room 520, 17 Xuzhou Road
Taipei 10055
Taiwan
Tel: 886-2-23920460, Fax: 886-2-23920456
E-mail: weichu@ntu.edu.tw